Author's response to reviews

Title: Does BMI generated by self-reported height and weight measure up in older adults from middle-income countries? Results from the Study on global AGEing and adult health (SAGE).

Authors:

Theresa E. Gildner (tgildner@uoregon.edu)
Tyler M. Barrett (tbarrett@uoregon.edu)
Melissa A. Liebert (liebert@uoregon.edu)
Paul Kowal (kowalp@who.int)
J. Josh Snodgrass (jjosh@uoregon.edu)

Version: 2  Date: 6 May 2015

Author's response to reviews:

Line numbers have been added to the manuscript.